Author's response to reviews

Title: A novel deletion mutation in the TUSC3 gene in a consanguineous Pakistani family with autosomal recessive nonsyndromic intellectual disability

Authors:

Muzammil Ahmad Khan (muzammilgandapur1983@gmail.com)
Muhammad Arshad Rafiq (Muhammad_Rafiq@camh.net)
Abdul Noor (Abdul_Noor@camh.net)
Nadir Ali (nadiralikerio@yahoo.com)
Ghazanfar Ali (ali.phd.qau@gmail.com)
John B Vincent (John_Vincent@camh.net)
Muhammad Ansar (ansar@qau.edu.pk)

Version: 2 Date: 14 January 2011

Author's response to reviews: see over
Editor
BMC Medical Genetics

Dear Sir,
Here we are submitting our revised manuscript, “A novel deletion mutation in the TUSC3 gene in a consanguineous Pakistani family with autosomal recessive nonsyndromic intellectual disability” by Khan et al. as research article for publication in BMC Medical Genetics.

We certify that the data in the manuscript is original and presently not submitted to any other journal for publication. It is further certified that current study conforms to Helsinki Declaration and local legislation. Written informed consents were obtained from parents/guardians of patients to publish their photographs.

Thank you for your time and efforts to review our manuscript for publication in BMC Medical Genetics.

Sincerely,

Muhammad Ansar, PhD
Assistant Professor
Department of Biochemistry
Faculty of Biological Sciences
Quaid-i-Azam University, Islamabad
PAKISTAN
Tel. No. 92-51-90643146
E.mail: ansar@qau.edu.pk
mansar_76@yahoo.com